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Weekly Status Week of June 12 – June 18, 2008
Day of Year 164 - 170
Mission Day 2768 - 2774
Earth Observing-One (EO-1) General

There were 98 Data Collection Events (DCEs) scheduled this week.  
INSTRUMENTS
All instruments operated nominally this week.
Instrument decontamination cycle:
· ALI out-gassing was performed on June 18 (170) from 02:07z to 13:59z.
· HIS de-icing was performed on June 18 (170) from 01:59z to 13:45z.
North Pole Special Collects:
· First DCE was executed on Monday June 9, 2008 with a scene start time of 13:11:05z.

· This image appeared to be mostly cloud covered.

· Second DCE was executed on Thursday June 12, 2008 with a scene start time of 13:28:10z. 

· First analysis from the Mission Science Office indicates that this DCE was largely free of clouds.
· The MSO is requesting additional North Pole collects.  The FOT is analyzing how to make these DCE’s more routine.
EO-1 Spacecraft Subsystems
Command and Data Handling (C&DH)

Continued to experience problems playing back engineering data from its Solid-State-Recorders.
Technology Activities

In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

Sensor Web & Virtual Observatory Demonstrations
A meeting was held with the Mission Science Office (MSO) to discuss new product development and campaigns to validate those products.  Also, there were discussed tasking priorities between sensor web targets and science long term plan (LTP) targets.  Stu Frye and Pat Cappelaere have developed documentation that describes the integration of a sensor web tasking queue with the baseline LTP schedule and have distributed the documentation to the sensor web team and the MSO.
The weekly teleconference between the Advanced Information System Technology (AIST) Sensor Web collaborators was held on Thursday, June 12.  The following items of interest were discussed:
· The cloud web service interface being developed by Draper Lab using Air Force data was discussed.  One item of interest was to make the Draper Lab service provide an interface compatible with the cloud web service that NOAA provides in addition to providing an OGC compliant interface.
· The team decided to submit a Notice of Intent (NOI) to the ROSES Decision proposal call.  The NOI was submitted on June 16.
· The team intends to submit two proposals to the ROSES AIST for sensor web activities for wildfires in the western US and floods in Central American and Africa.

· We ordered a Level 1R ALI scene from the USGS to compare against a JPL Level 1R version of the same scene to resolve an apparent difference in one of the four image panels.  JPL engineers are investigating the source of the difference and will report on their findings.

· Dan Mandl requested a Quickbird image of Myanmar to demonstrate to the IFRC flood intelligence from multiple sources.  
Autonomous Sciencecraft Experiment (ASE)

The ASE controlled EO-1 all week.
EO-1 MISSION OPERATIONS CENTER
Real-Time
Spacecraft state of health is currently nominal.
Solar array characterization test performed on June 18 (170) at 12:19z.
Mission Planning
Mission planning functions are being performed using older versions of MOPPS on Bullseye system.

ASPEN upgrade:

· No change since last report.

Flight Dynamics
North Pole collect occurred as noted stated above.

Data Analysis and Trending

No change from last report.

System Administration
Systems upgrades and IT security requirements:

No change from last report.

Operational Enhancements Requests and Discrepancy Reports
No change from last report.
GROUND AND SPACE NETWORK

Station Downtime
No major problems to report.
Operational Discrepancies
No major problems to report.
Support Enhancements  
Certification of PF-2:
· FOT has performed two S-Band contacts utilizing PF-2 systems.
· First contact was a 13 min support with Acquisition of Signal (AOS) at approximately 19:20z and Loss of Signal (LOS) at approximately 19:33z on June 16 (168).
· Real-time automation on the T&C software has been configured to handle PF-2 in the execution schedule.
· During this contact telemetry was captured and observed successfully without any sequence breaks or drop-outs.  Also during this support commands were radiated and accepted by the spacecraft successfully.

· Second contact was a 12 min support with AOS at approximately 18:21z and LOS at approximately 18:33z on June 17 (169).

· During this contact, FOT did not have commanding of the spacecraft until Universal Space Network (USN) was notified through the black-phone to perform a re-sweep.

· Once the re-sweep of the antenna was complete FOT was able to radiate commands to the spacecraft. 

· FOT will request an anomaly report from USN and the PF-2 systems to troubleshoot this event.

· FOT will perform one more S-Band contact and execute three X-Band engineering contacts to fully certify the PF-2 systems for both S-Band and X-Band operations.

WS1 antenna testing:
No change from last report.

Upcoming Events
Lunar Calibration:

· An all instrument nominal lunar calibration will be performed on June 19, 2008 during the 07:55z umbra.
· A modified lunar calibration will be performed three orbits later.
· Once the modified lunar calibration has been validated, nominal calibrations will be discontinued.
Imagery Status
Scenes and Engineering Cals planned for week of June 12 – June 18, 2008              98
Total scenes and engineering calibrations planned for entire mission – approximately 40,760
Total Scenes:  ALI scenes in the Level 0 archive            36,638 (as of June 18, 2008)

                        Hyperion scenes in the Level 0 archive    36,387
Publications and Presentations Status (as of 12-05-07)
404 publications 

284 external presentations

53 articles and press releases
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